Author's response to reviews

Title: Effect of needle puncture and electro-acupuncture on mucociliary clearance in anesthetized quails

Authors:

Shusheng Tai (stai@ualberta.ca)
Jiulin Wang (WangJ@MacEwan.ca)
Feng Sun (fs1@ualberta.ca)
Stevenson Xutian (xutian1@netzero.net)
Malcolm King (malcolm.king@ualberta.ca)

Version: 2 Date: 22 June 2005

Author’s response to reviews:

We would like to add more peer reviewers so that you may find enough reviewers for the manuscript.

Chun-su Yuan, cyuan@uchicago.edu, Medicine, University of Chicago

Bruce Rubin, brubin@wfubmc.edu, Pediatrics, Wake Forest University

David Eidelman, david.eidelman@mcgill.ca, Respiratory Physiology, McGill University

Terry Dwyer, tdwyer@physiology.umsmed.edu, Physiology & Biophysics, University of Mississippi

Haoqing Tai, bllx@njutcm.edu.cn, Anatomy, Nanjing University of Traditional Chinese Medicine

Moriuchi Hiroshi, moriuchi@gpo.kumamoto-u.ac.jp, Clinical chemistry and informatics, Kumamoto University

John Fisher, fisherjt@post.queensu.ca, physiology, Queen's University